Author's response to reviews

Title: Improving recruitment to a study of telehealth management for long-term conditions in primary care: two embedded, randomised controlled trials of optimised patient information materials

Authors:

Mei-See Man (mei-see.man@bristol.ac.uk)
Jo Rick (jo.rick@manchester.ac.uk)

Version: 3 Date: 24 June 2015

Author's response to reviews:

Dear Editor,

As requested we have made the following changes to the manuscript:

1. email addresses have been added for Mei-See Man and Jo Rick
2. The figure listing has been removed from the additional files list and a figures list created in the appropriate place in the manuscript
3. a postal address has been added to the corresponding author details on the cover page.

Please note that we have also changed the corresponding author to Peter Bower (MRC START PI).

Many thanks,

Jo Rick